Author's response to reviews

Title: The relationship between body composition and physical fitness in 14 year-old adolescents residing within the Tlokwe Local Municipality, South Africa: The PAHL-Study

Authors:

Makama A Monyeki (andries.monyeki@nwu.ac.za)
Rik Neetens (rikneetens@gmail.com)
Sarah J Moss (hanlie.moss@nwu.ac.za)
Jos WR Twisk (jwr.twisk@vumc.nl)

Version: 5 Date: 13 April 2012

Author's response to reviews: see over
RE: SUBMISSION OF REVISED RESEARCH MANUSCRIPT (MS: 1961608540623743)
FOR PUBLICATION

Enclosed please find a copy a revised manuscript titled: “The relationship of body composition with physical fitness in the 14 years old adolescents residing within the Tlokwe Local Municipality, South Africa: The PAHL-Study.” The manuscript is being revised in accordance with the reviewers comments (see included responses and the manuscript). This manuscript is an original and it was never been published in any journal before.

I also wanted to inform you that manuscript is been language edited by an English language specialist at our North-West University.

We also like to take this opportunity to thank the BMC Public Health editor’s willingness and time to review our manuscript, it is highly appreciated.

Hoping to hear from you.

Thank you

Yours sincerely,

Prof Andries Monyeki